<mark>_</mark>



Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2024GL109717

Key Points:

- Direct use of satellite laser ranging (SLR)-derived low-degree gravity field yields significantly underestimated global ocean mass rate due to severe signal leakage
- The leakage-corrected SLR gravity field provides a realistic estimate of global ocean mass rate that agrees remarkably well with Gravity Recovery and Climate Experiment
- Forward modeling is more effective than the buffer zone technique in leakage correction for low-degree gravity field

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

Y. Nie, yufeng.nie@polyu.edu.hk

Citation:

Nie, Y., Chen, J., & Peng, D. (2024). Global ocean mass change estimation using low-degree gravity field from satellite laser ranging. *Geophysical Research Letters*, *51*, e2024GL109717. https://doi.org/10.1029/2024GL109717

Received 9 APR 2024 Accepted 24 MAY 2024

Author Contributions:

Conceptualization: Yufeng Nie Data curation: Yufeng Nie Formal analysis: Yufeng Nie Funding acquisition: Jianli Chen Investigation: Yufeng Nie, Jianli Chen, Dongju Peng Methodology: Yufeng Nie, Jianli Chen Project administration: Jianli Chen Resources: Jianli Chen Software: Yufeng Nie Supervision: Jianli Chen Validation: Yufeng Nie Visualization: Yufeng Nie Writing – original draft: Yufeng Nie

© 2024. The Author(s). This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Global Ocean Mass Change Estimation Using Low-Degree Gravity Field From Satellite Laser Ranging

Yufeng Nie¹, Jianli Chen^{1,2,3}, and Dongju Peng^{1,3}

¹Department of Land Surveying and Geo-Informatics, The Hong Kong Polytechnic University, Hong Kong, China, ²Research Institute for Land and Space, The Hong Kong Polytechnic University, Hong Kong, China, ³Shenzhen Research Institute, The Hong Kong Polytechnic University, Shenzhen, China

Abstract Satellite laser ranging (SLR) is a well-established geodetic technique for measuring the lowdegree time-variable gravity field for decades. However, its application in mass change estimation is limited by low spatial resolution, even for global mean ocean mass (GMOM) change which represents one of the largest spatial scales. After successfully correcting for signal leakage, for the first time, we can infer realistic GMOM changes using SLR-derived gravity fields up to only degree and order 5. Our leakage-corrected SLR GMOM estimates are compared with those from the Gravity Recovery and Climate Experiment (GRACE) for the period 2005 to 2015. Our results show that the GMOM rate estimates from SLR are in remarkable agreement with those from GRACE, at 2.23 versus 2.28 mm/year, respectively. This proof-of-concept study opens the possibility of directly quantifying GMOM change using SLR data prior to the GRACE era.

Plain Language Summary The global mean ocean mass (GMOM) is a significant contributor to the current rise of global sea level. Satellite laser ranging (SLR) has been a proven geodetic technique that can determine low-degree time-varying gravity fields for over three decades. It is possible to derive the GMOM using only long-wavelength gravity field information from SLR because of its global nature. However, the coarse resolution of SLR gravity field makes it unable to differentiate between mass changes from land and oceans. A large portion of land signals leak into the adjacent oceans, leading to significantly underestimated GMOM change rates. To solve this problem, we incorporate geographic knowledge of ocean and land boundaries into the GMOM estimation process. This can help restore the lost land signals and results in an improved GMOM estimate. We compare the SLR estimates with the more reliable results from GRACE, which shows that the GMOM rate from SLR, after correcting for leakage, agrees remarkably well with that from GRACE. This result is encouraging because it allows for the direct quantification of GMOM change for years before 2002, which is critical for understanding long-term global sea level rise.

1. Introduction

Present-day global mean sea level (GMSL) change includes two main components: the change in seawater density due to temperature and salinity variations (the steric component), and the change in global mean ocean mass (GMOM) due to net freshwater exchange between the land and oceans (the barystatic component) (Cazenave & Moreira, 2022; Gregory et al., 2019; WCRP Global Sea Level Budget Group, 2018). Since 1992, satellite altimetry has provided accurate and continuous measurements of global and regional sea level change (Cazenave et al., 2018). The steric component of GMSL can be derived from ocean reanalysis models (Storto & Yang, 2024; Storto et al., 2015) or in situ hydrographic measurements collected from global networks such as the Argo array (Johnson et al., 2022). For GMOM change, we can infer it indirectly by summing all land mass changes in polar ice sheets, mountain glaciers, and terrestrial water storage (TWS), known as the ocean mass budget approach (Barnoud et al., 2023; Chambers et al., 2016; Dieng et al., 2017; Horwath et al., 2022; Llovel et al., 2023; WCRP Global Sea Level Budget Group, 2018). Alternatively, direct quantification has been available since the launch of the Gravity Recovery and Climate Experiment (GRACE) mission in 2002 (Tapley et al., 2004), which routinely produces high-precision, high-resolution monthly gravity field models that are widely used for GMOM estimation (Chambers et al., 2004; Chen et al., 2013, 2018; Dobslaw et al., 2020; Jeon et al., 2021; Kim et al., 2019; Rietbroek et al., 2016; Uebbing et al., 2019; Yi et al., 2015, 2017). Ideally, the sum of the barystatic and steric components should agree with the altimetry-based GMSL change within the data uncertainties, indicating that the GMSL budget is closed. It is generally accepted that GMSL closure is achieved between 2005 and 2015, when the three complementary observational techniques (GRACE, Argo, and satellite



Writing – review & editing: Yufeng Nie, Jianli Chen, Dongju Peng altimetry) are all in nominal operation, providing data with sufficient accuracy and coverage (Chen et al., 2018; WCRP Global Sea Level Budget Group, 2018). There are notably large discrepancies between GRACE (plus GRACE Follow-On (FO), launched in 2018) observed GMOM change and estimates from satellite altimetry and Argo, which were likely attributed to errors in Argo and GRACE/GRACE-FO estimates (Barnoud et al., 2021, 2023; Chen et al., 2020; Horwath et al., 2022; Johnson et al., 2023).

Prior to the GRACE era, satellite laser ranging (SLR) played a key role in determining the low-degree timevariable gravity field, starting with the launch of LAGEOS-1 and Starlette in the 1970s (Pearlman et al., 2019). Since the 1990s, with several dedicated SLR satellites in orbit, the spherical harmonic (SH) expansion of gravity field models derived from SLR can reach up to degree and order (d/o) 5, corresponding to ~4,000 km spatial resolution (half wavelength) (Cheng et al., 1997, 2011; Löcher & Kusche, 2020; Loomis et al., 2019, 2020; Matsuo et al., 2013; Sośnica et al., 2015). Numerous studies have focused on analyzing the temporal variations of these low-degree coefficients, particularly the zonal term J2 that reflects the dynamic oblateness of the Earth. Specifically, various geophysical models are used to explain the observed changes, which are associated with large-scale mass redistribution in the Earth system, from seasonal to long-term scales (Chao et al., 2020; Cheng & Ries, 2018; Cheng & Tapley, 2004; Cheng et al., 2013; Cox & Chao, 2002; Nerem & Wahr, 2011; Nerem et al., 2000; Yoder et al., 1983). However, the inverse process, that is, using the SLRderived gravity field to directly infer regional mass change, is challenging due to its very limited spatial resolution, even for regions as large as the Greenland ice sheet (GrIS) and the Antarctic ice sheet (AIS) (Bonin et al., 2018; Matsuo et al., 2013; Meyer et al., 2019). To improve estimates of regional mass change, it is often necessary to incorporate a prior information using various methods: spatially via the least-squares Mascon approach (Bonin et al., 2018), spectrally via the scaling factor technique (Gałdyn et al., 2024), or spatiotemporally via empirical orthogonal function projection (Löcher & Kusche, 2020; Talpe et al., 2017). While these approaches improve interpretability for regional studies, they reduce the independency of SLR from other observational techniques.

The situation is different for the global ocean, which covers about 70% of the Earth's surface, a spatial scale large enough to allow, in principle, the use of the low-degree gravity field for direct mass change estimation. In practice, however, there is significant signal leakage between land and oceans due to the rather limited spatial resolution. Without effectively addressing signal leakage, it is unlikely to derive realistic GMOM estimates from the low-degree SLR solutions. This also explains why SLR is rarely, if not impossible, used to directly infer GMOM changes, even though the global nature of GMOM implies that only long-wavelength information is sufficient. We address this problem by correcting for signal leakage using the global forward modeling (FM) technique (Chen et al., 2015). The leakage-corrected SLR solutions are then used to estimate the GMOM change and are validated with GRACE estimates. To our knowledge, this is the first study to use an independent low-degree (5 d/o) SLR gravity field for GMOM estimation. Our proof-of-concept study opens the possibility of directly estimating GMOM change using SLR for the pre-GRACE era.

2. Data and Methods

2.1. Gravity Field Models

For SLR gravity field models, we use two operational data sets: one from the Center for Space Research (CSR) at the University of Texas at Austin (Cheng & Ries, 2017) and the other from NASA Goddard Space Flight Center (GSFC) (Loomis et al., 2019, 2020). Both solutions are estimated up to 5 d/o (plus additional SH coefficients C_{61} and S_{61}) based on tracking data from multiple SLR satellites, including LAGEOS-1/2, Stella, Starlette, Ajisai and LARES (see Table S1 in Supporting Information S1 for information on satellites). The GSFC solution additionally includes the Larets satellite (Loomis et al., 2020). The CSR solution is sampled monthly and spans from 2002 to the present, while the GSFC solution is weekly starting from 2000. We resample the weekly GSFC solutions to their monthly averages. To reduce data noise, we use the ensemble mean of the two data sets. For the GRACE gravity field models, we use the Level-2 SH solutions up to 60 d/o provided by CSR (Release 6.0). As recommended, the J_2 coefficients in GRACE solutions are replaced by those derived from SLR given in Technical Note TN-14 (Loomis et al., 2019). We note that GRACE Mascon solutions are less affected by signal leakage, whereas GMOM estimates from either SH or Mascon solutions are generally consistent (Barnoud et al., 2023; Chen et al., 2020; Horwath et al., 2022). The time span for our investigations is January 2005 to December 2015, and the static gravity field is removed using the GOCO06s model to derive anomaly fields (Kvas et al., 2021). We also remove the glacial isostatic adjustment signals using the ICE6G-D model (Peltier et al., 2018). The non-tidal dynamic ocean signals, represented as the GAB component of the Atmosphere and Ocean De-Aliasing Level-1B (AOD1B) product, are not restored because they have minimal impact on the GMOM estimates since Release 06 (Dobslaw et al., 2017, 2020). For geocenter motions (degree-1 spherical harmonics), there are notable inconsistencies between estimates based on different techniques, such as GRACE (plus ocean models), SLR, and Global Navigation Satellite Systems, which contribute to one of the largest uncertainties in GMOM estimates (Blazquez et al., 2018; Chen et al., 2018; Horwath et al., 2022; Kim et al., 2022). Furthermore, the widely used GRACE-based geocenter estimates (Sun et al., 2016; Swenson et al., 2008) are not available in the pre-GRACE era, while there is still room for improvement in the SLR-based geocenter estimates (Cheng, 2024). Therefore, like previous studies (Bonin et al., 2018; Löcher & Kusche, 2020; Meyer et al., 2019; Talpe et al., 2017), we omit them for GRACE and SLR solutions, which has a common impact on both and is subject to further investigation. The anomaly SH coefficients are transformed into gridded equivalent water heights (Wahr et al., 1998), which are used for GMOM estimation.

2.2. Leakage Correction

Due to the truncated SH expansion, satellite-based gravity field models are generally limited in their ability to separate signals from different source locations, leading to so-called signal leakage in regional mass change estimation (Swenson & Wahr, 2002). Signal leakage is further exacerbated by the spatial filtering required to suppress high-frequency noise. For the global ocean, signal leakage mainly occurs along coastlines, where much stronger land signals, for example, from melting ice sheets, leak into the surrounding oceans, leading to a general underestimation of ocean mass (Chen et al., 2013). A widely used method for leakage correction in GMOM estimation is to apply a buffer zone that extends the coastline hundreds of kilometers into the ocean to mask out areas near land, for example, 300–500 km typically for GRACE (Barnoud et al., 2023; Chambers et al., 2007, 2016; Chen et al., 2018; Uebbing et al., 2019). However, for the SLR solutions of only 5 d/o, the leakage problem is much more severe than for GRACE (60 d/o), indicating that an extremely large buffer zone is needed.

In this study, we correct the signal leakage by FM, which has been proven effective in various regional GRACE applications (Chen et al., 2007, 2009; Jiao et al., 2022; Long et al., 2015). For GMOM estimation, only the geographic knowledge of land and ocean boundaries is required (Chen et al., 2013; Jeon et al., 2021; Kim et al., 2019; WCRP Global Sea Level Budget Group, 2018; Yi et al., 2015). This is an important feature of our leakage correction method, which does not use GRACE information to spatially or temporally constrain the mass change patterns of low-resolution SLR gravity field. By assuming that the signals come mainly from land, the global FM iteratively updates the land signals to match the original "apparent" model when applying the same truncation and filtering process. A 4,000 km Gaussian filter is applied to suppress noise in the SLR gravity coefficients (Matsuo et al., 2013). The ocean mass is simultaneously adjusted for the global mass conservation at each iteration. When converged, it produces the leakage-corrected land mass distribution as well as the associated ocean mass; the latter, when integrated over the global ocean, naturally provides the desired GMOM estimate. We outline the processing steps of the FM for GMOM estimation in Text S1 in Supporting Information S1, and further details can be found in Chen et al. (2013, 2015). Moreover, if the focus is regional instead of global mean ocean mass changes, sea-level fingerprints can be inferred from the leakage-corrected land mass by solving the sea-level equation (Adhikari et al., 2019; Farrell & Clark, 1976; Hsu & Velicogna, 2017; Riva et al., 2010).

3. Results and Discussion

Figure 1a shows the GMOM change time series derived from GRACE (60 d/o) and SLR (5 d/o) from 2005 to 2015. The leakage-corrected GRACE solution using global FM is used as the benchmark for comparison. Both SLR solutions, with and without leakage correction, are computed for comparison. All solutions show clear seasonal signals, even for the raw SLR estimate without leakage correction, that is, SLR (w/o FM). However, the long-term change is significantly underestimated in the raw SLR solution. This is expected because the GMOM rate of change is primarily driven by changes in land ice mass, including polar ice sheets and mountain glaciers, which typically occur along coastlines. The low resolution of the SLR gravity field makes it difficult to accurately locate these ice-melt signals, which leak into adjacent oceans and are misinterpreted as changes in ocean mass. As





Figure 1. (a) Global mean ocean mass change from the Gravity Recovery and Climate Experiment (60 d/o) and satellite laser ranging (5 d/o) gravity fields with and without (w/o) leakage correction using global forward modeling. Panel (b) same as (a), but with the seasonal terms (annual and semi-annual) removed, and the dashed lines represent the estimated trends from least-squares fits.

the FM technique is particularly effective in recovering land ice signals, the leakage-corrected SLR solution is in remarkably good agreement with GRACE for long-term signals.

Figure 1b shows the leakage-corrected ocean mass variations (annual and semi-annual terms removed) together with the linear trends (dashed lines) obtained by unweighted least-squares fits. On the one hand, the leakage-corrected SLR solution not only provides consistent trend estimates with GRACE, but also captures interannual signals mainly related to the El Nino-Southern Oscillation events, such as the well-observed ocean mass decrease in 2011 due to La Niña (Boening et al., 2012). On the other hand, it has more variability than the other two solutions. The observational data for SLR are much sparser and more inhomogeneous than those for GRACE, which results in higher noise and correlations in the estimated gravity field parameters. The noise is further amplified by the leakage correction using FM, which is essentially an inversion process. Nevertheless, the trend estimation is less affected in our case because the 10-year period is not too short to distinguish between trend and interannual variability (Cazenave et al., 2014; Moreira et al., 2021).

Figure 2 shows the gridded mass change rates estimated from GRACE and SLR gravity field solutions. Regional ice melt patterns in the GrIS, AIS, and mountain glaciers are well observed by GRACE and dominate the current ocean mass increase (Bamber et al., 2018). The raw SLR gravity field captures some of these signals, as seen near GrIS and the West AIS, but they are scattered far from the source with significantly reduced magnitudes. In contrast, the global FM successfully relocates the leaked land signals and largely restores their magnitudes, which is critical for deriving a reliable GMOM rate. In addition, the leakage-corrected SLR solution can qualitatively distinguish the mass gain in the East AIS from the mass loss in the West AIS. However, given its native spatial resolution of only 5 d/o, we cannot expect it to reveal regional details like GRACE, for example, the distinct patterns between inland and coastal regions of the GrIS are not separated. However, this will have minimal impact on our GMOM estimate, since at such a global scale only the total amount of land-ocean mass exchange is relevant.





Figure 2. Gridded trends of mass changes (in terms of equivalent water heights) estimated from Gravity Recovery and Climate Experiment (60 d/o) and satellite laser ranging (SLR) (5 d/o) after leakage correction using the global forward modeling (FM). The raw SLR solution without leakage correction, that is, SLR (w/o FM), is also shown. A uniform layer of water mass is assigned over the oceans for global mass conservation in the FM.

Figure 3 shows the gridded annual amplitudes of the three solutions, which mainly show seasonal variations in TWS. As observed by GRACE, large annual signals are present in large river basins, particularly in tropical regions (Scanlon et al., 2019). Changes in TWS are the main contributor to the seasonal variations in GMOM, as the atmospheric and dynamical ocean signals are removed during the gravity field recovery process (Willis et al., 2008). Again, the leakage-corrected SLR solution can generally recover annual signals at the continental scale.

Table 1 lists the estimated trends and seasonal amplitudes and phases of GMOM changes from 2005 to 2015, as well as their formal one-sigma uncertainties obtained from least-squares fits. In addition, SLR estimates using different leakage-correction buffers are presented for comparison. The Gaussian filter is not applied for the case using buffer zone. For linear rates, the leakage-corrected SLR solution (2.23 mm/year) agrees remarkably well with GRACE (2.28 mm/year), mainly due to the effectiveness of the FM in recovering land ice mass change signals, as shown in Figure 2. Without any correction for leakage, the 5 d/o SLR solution obviously cannot be used to derive the GMOM rate, as it gives an estimate of only 0.32 mm/year. The use of a buffer zone can recover some of the long-term signals, and the estimated rates are 0.98, 1.21, and 1.27 mm/year for buffer zones of 500, 800, and 1,000 km, respectively, gradually increasing with the size of the buffer zone. However, even for a 1,000 km buffer zone, the estimated rate is still about half that of GRACE and SLR when using FM. Given the spatial resolution of only 5 d/o for the SLR gravity field, the widely used buffer zone technique in GRACE cannot provide realistic GMOM rate estimates.

For the annual signals, all SLR solutions are in good phase agreement with GRACE, which benefits from the combination of the two SLR solutions from CSR and GSFC. As shown in Fig. S1 in Supporting Information S1, the phasor plot indicates that the combination of the two SLR solutions brings the annual phase closer to that of GRACE. The amplitudes of all SLR solutions in Table 1 are conservative, while FM gives the largest estimate of 9.11 mm, but is still about 10% smaller than the 10.14 mm of GRACE. The individual GMOM estimates of CSR













(cm)

and GSFC in Table S2 in Supporting Information S1 show that the GSFC solution has a smaller annual amplitude than CSR, but a larger semi-annual amplitude. This may be related to their specific SLR data processing strategies, for example, background models and/or parameterizations, in gravity field modeling (Cheng & Ries, 2023), which requires further investigation. In contrast to rate estimation, the buffer zone can bring back the annual signals, for example, 8.91 mm for the 1,000 km case. This can be explained by the fact that most basins with significant seasonal signals, such as the Amazon, are located inland (see Figure 3). Therefore, the use of a sufficiently large buffer zone around land can account for the leakage signals. However, long-term mass changes occur mainly along narrow coastlines, especially for melting ice sheets. As a result, they naturally leak further out into the oceans than annual signals from inland areas and require an extremely large buffer zone to be effectively

Table 1

Linear Trends and Seasonal Signals of Global Mean Ocean Mass (GMOM) Changes From Gravity Recovery and Climate Experiment (60 d/o) and Satellite Laser Ranging (5 d/o) for the Period 2005–2015

	Trend (mm/year)	Annual		Semiannual	
GMOM		Amplitude (mm)	Phase (deg)	Amplitude (mm)	Phase (deg)
GRACE (FM)	2.28 ± 0.07	10.14 ± 0.29	273 ± 2	1.01 ± 0.29	39 ± 17
SLR (FM)	2.23 ± 0.14	9.11 ± 0.63	275 ± 4	0.99 ± 0.63	32 ± 36
SLR (w/o FM, no BUF)	0.32 ± 0.06	5.71 ± 0.26	279 ± 3	0.31 ± 0.26	34 ± 48
SLR (BUF 500 km)	0.98 ± 0.10	7.85 ± 0.43	277 ± 3	0.35 ± 0.43	27 ± 70
SLR (BUF 800 km)	1.21 ± 0.11	8.50 ± 0.50	278 ± 3	0.37 ± 0.50	17 ± 78
SLR (BUF 1,000 km)	1.27 ± 0.12	8.91 ± 0.54	279 ± 3	0.39 ± 0.54	12 ± 79

Note. FM means the global forward modeling for leakage correction, and BUF means the buffer zone.



accounted for. For the semi-annual signals, the use of FM allows very close SLR estimates to GRACE, while setting different buffer zones cannot resolve the amplitude underestimation.

4. Concluding Remarks

We investigate the potential of using the low-degree SLR gravity field for GMOM estimation. Given its very limited spatial resolution of 5 d/o, direct use results in significant underestimation of GMOM change rates and seasonal signals due to strong signal leakage between land and oceans. We apply global FM to correct for leakage by incorporating geographic information on land-ocean boundaries. The leakage-corrected SLR gravity field is used for GMOM estimation and compared with GRACE (60 d/o) over the period 2005 to 2015, during which GRACE estimates can be considered reliable. Without leakage correction, the SLR gives a GMOM rate of only 0.32 mm/year, which is significantly lower than the GRACE reference of 2.28 mm/year. Using global FM, the estimated rate is improved to 2.23 mm/year, which agrees remarkably well with GRACE, though the formal uncertainty is larger than that of GRACE. The finding that a low-degree gravity field is capable of deriving realistic GMOM trend may initially seem surprising. However, it is important to note that the GMOM represents one of the largest spatial scales, so long-wavelength gravity change information is adequate provided that signal leakages along the coastlines are sufficiently corrected. By assuming most of signals come from the land, which is the case for long-term trends in general, the FM successfully restores the land signals, especially polar ice mass changes that drive the GMOM trend. This explains the good agreement between the 5 d/o SLR and 60 d/o GRACE solutions with FM. For regional applications, however, high-degree information is required for signal separation even with FM. Establishing a buffer zone up to 1,000 km from the coastlines cannot resolve the leakage as it gives a rate of ~ 1.27 mm/year, which is still significantly underestimated. This indicates that the buffer zone technique is less effective than the FM technique given the coarse spatial resolution of the continental-scale SLR gravity field. For seasonal variations, both buffer zone and FM techniques can recover the signals, but the latter provides a better estimate of the semi-annual amplitude. Our proof-of-concept study gives encouraging results based on the 5 d/o SLR gravity field for GMOM estimation, which was previously considered infeasible or not even discussed. This opens up a potential new scientific application aspect of the long-standing and wellestablished SLR technique. In the future, when operational SLR gravity solutions are available for the pre-GRACE period, a more comprehensive analysis of the global ocean mass budget can be performed by reconciling direct SLR estimates with those synthesized from various mass balance data sets of ice sheets, glaciers, and TWS.

Data Availability Statement

The SLR gravity field models are available from https://ftp.csr.utexas.edu/pub/slr/degree_5/CSR_Monthly_5x5_ Gravity_Harmonics.txt (CSR) and https://earth.gsfc.nasa.gov/sites/default/files/geo/slr-weekly/gsfc_slr_ 5x5c61s61.txt (GSFC). GRACE CSR RL06 models are from GRACE (2018). The J2 estimates from TN-14 are provided at https://archive.podaac.earthdata.nasa.gov/podaac-ops-cumulus-docs/gracefo/open/docs/TN-14_ C30_C20_GSFC_SLR.txt.

Acknowledgments

The authors are grateful to the two anonymous reviewers for their comprehensive and insightful comments, which have led to improved presentation of the results. This work was supported by the NSFC National Key Project (42394132, 42394131), Hong Kong RGC Collaborative Research Fund (C5013-23G) and PolyU SHS and LSGI Internal Research Funds (Project IDs: P0042322 and P0041486).

References

Adhikari, S., Ivins, E. R., Frederikse, T., Landerer, F. W., & Caron, L. (2019). Sea-level fingerprints emergent from GRACE mission data. *Earth System Science Data*, 11(2), 629–646. https://doi.org/10.5194/essd-11-629-2019

Bamber, J. L., Westaway, R. M., Marzeion, B., & Wouters, B. (2018). The land ice contribution to sea level during the satellite era. Environmental Research Letters, 13(6), 063008. https://doi.org/10.1088/1748-9326/aac2f0

Barnoud, A., Pfeffer, J., Cazenave, A., Fraudeau, R., Rousseau, V., & Ablain, M. (2023). Revisiting the global mean ocean mass budget over 2005–2020. Ocean Science, 19(2), 321–334. https://doi.org/10.5194/os-19-321-2023

Barnoud, A., Pfeffer, J., Guérou, A., Frery, M. L., Siméon, M., Cazenave, A., et al. (2021). Contributions of altimetry and Argo to non-closure of the global mean Sea Level budget since 2016. *Geophysical Research Letters*, 48(14), e2021GL092824. https://doi.org/10.1029/2021gl092824

Blazquez, A., Meyssignac, B., Lemoine, J. M., Berthier, E., Ribes, A., & Cazenave, A. (2018). Exploring the uncertainty in GRACE estimates of the mass redistributions at the Earth surface: Implications for the global water and sea level budgets. *Geophysical Journal International*, 215(1), 415–430. https://doi.org/10.1093/gji/ggy293

Boening, C., Willis, J. K., Landerer, F. W., Nerem, R. S., & Fasullo, J. (2012). The 2011 La Niña: So strong, the oceans fell. Geophysical Research Letters, 39(19). https://doi.org/10.1029/2012gl053055

Bonin, J. A., Chambers, D. P., & Cheng, M. (2018). Using satellite laser ranging to measure ice mass change in Greenland and Antarctica. *The Cryosphere*, 12(1), 71–79. https://doi.org/10.5194/tc-12-71-2018

Cazenave, A., Dieng, H.-B., Meyssignac, B., von Schuckmann, K., Decharme, B., & Berthier, E. (2014). The rate of sea-level rise. *Nature Climate Change*, 4(5), 358–361. https://doi.org/10.1038/nclimate2159

Cazenave, A., & Moreira, L. (2022). Contemporary sea-Level changes from global to local scales: A review. *Proceedings of the Royal Society A: Mathematical, Physical & Engineering Sciences,* 478(2261), 20220049. https://doi.org/10.1098/rspa.2022.0049

- Cazenave, A., Palanisamy, H., & Ablain, M. (2018). Contemporary sea level changes from satellite altimetry: What have we learned? What are the new challenges? Advances in Space Research, 62(7), 1639–1653. https://doi.org/10.1016/j.asr.2018.07.017
- Chambers, D. P., Cazenave, A., Champollion, N., Dieng, H., Llovel, W., Forsberg, R., et al. (2016). Evaluation of the global mean Sea Level budget between 1993 and 2014. Surveys in Geophysics, 38(1), 309–327. https://doi.org/10.1007/s10712-016-9381-3
- Chambers, D. P., Tamisiea, M. E., Nerem, R. S., & Ries, J. C. (2007). Effects of ice melting on GRACE observations of ocean mass trends. *Geophysical Research Letters*, 34(5), L05610. https://doi.org/10.1029/2006gl029171
- Chambers, D. P., Wahr, J., & Nerem, R. S. (2004). Preliminary observations of global ocean mass variations with GRACE. *Geophysical Research Letters*, *31*(13), L13310. https://doi.org/10.1029/2004gl020461
- Chao, B. F., Yu, Y., & Chung, C. H. (2020). Variation of Earth's oblateness J2 on interannual-to-decadal timescales. Journal of Geophysical Research: Solid Earth, 125(6), e2020JB019421. https://doi.org/10.1029/2020jb019421
- Chen, J., Tapley, B., Save, H., Tamisiea, M. E., Bettadpur, S., & Ries, J. (2018). Quantification of Ocean Mass change using gravity recovery and climate experiment, satellite altimeter, and Argo floats observations. *Journal of Geophysical Research: Solid Earth*, 123(11), 10-212. https:// doi.org/10.1029/2018jb016095
- Chen, J., Tapley, B., Wilson, C., Cazenave, A., Seo, K. W., & Kim, J. S. (2020). Global ocean mass change from GRACE and GRACE follow-on and altimeter and Argo measurements. *Geophysical Research Letters*, 47(22), e2020GL090656. https://doi.org/10.1029/2020gl090656
- Chen, J. L., Wilson, C. R., Blankenship, D., & Tapley, B. D. (2009). Accelerated Antarctic ice loss from satellite gravity measurements. *Nature Geoscience*, 2(12), 859–862. https://doi.org/10.1038/ngeo694
- Chen, J. L., Wilson, C. R., Li, J., & Zhang, Z. (2015). Reducing leakage error in GRACE-observed long-term ice mass change: A case study in West Antarctica. Journal of Geodesy, 89(9), 925–940. https://doi.org/10.1007/s00190-015-0824-2
- Chen, J. L., Wilson, C. R., & Tapley, B. D. (2013). Contribution of ice sheet and mountain glacier melt to recent sea level rise. *Nature Geoscience*, 6(7), 549–552. https://doi.org/10.1038/ngeo1829
- Chen, J. L., Wilson, C. R., Tapley, B. D., Blankenship, D. D., & Ivins, E. R. (2007). Patagonia icefield melting observed by gravity recovery and climate experiment (GRACE). *Geophysical Research Letters*, 34(22), 22501. https://doi.org/10.1029/2007gl031871
- Cheng, M. (2024). An updated estimate of geocenter variation from analysis of SLR data. *Remote Sensing*, 16(7), 1189. https://doi.org/10.3390/rs16071189
- Cheng, M., & Ries, J. (2017). The unexpected signal in GRACE estimates of C20. Journal of Geodesy, 91(8), 897–914. https://doi.org/10.1007/ s00190-016-0995-5
- Cheng, M., & Ries, J. (2023). C20 and C30 variations from SLR for GRACE/GRACE-FO science applications. Journal of Geophysical Research: Solid Earth, 128(2), e2022JB025459. https://doi.org/10.1029/2022jb025459
- Cheng, M., & Ries, J. C. (2018). Decadal variation in Earth's oblateness (J2) from satellite laser ranging data. *Geophysical Journal International*, 212(2), 1218–1224. https://doi.org/10.1093/gji/ggx483
- Cheng, M., Ries, J. C., & Tapley, B. D. (2011). Variations of the Earth's figure axis from satellite laser ranging and GRACE. Journal of Geophysical Research, 116(B1), B01409. https://doi.org/10.1029/2010jb000850
- Cheng, M., & Tapley, B. D. (2004). Variations in the Earth's oblateness during the past 28 years. *Journal of Geophysical Research*, 109(B9), B09402. https://doi.org/10.1029/2004jb003028
- Cheng, M., Tapley, B. D., & Ries, J. C. (2013). Deceleration in the Earth's oblateness. Journal of Geophysical Research: Solid Earth, 118(2), 740–747. https://doi.org/10.1002/jgrb.50058
- Cheng, M. K., Shum, C. K., & Tapley, B. D. (1997). Determination of long-term changes in the Earth's gravity field from satellite laser ranging observations. Journal of Geophysical Research, 102(B10), 22377–22390. https://doi.org/10.1029/97jb01740
- Cox, C. M., & Chao, B. F. (2002). Detection of a large-scale mass redistribution in the terrestrial system since 1998. *Science*, 297(5582), 831–833. https://doi.org/10.1126/science.1072188
- Dieng, H. B., Cazenave, A., Meyssignac, B., & Ablain, M. (2017). New estimate of the current rate of sea level rise from a sea level budget approach. *Geophysical Research Letters*, 44(8), 3744–3751. https://doi.org/10.1002/2017g1073308
- Dobslaw, H., Bergmann-Wolf, I., Dill, R., Poropat, L., Thomas, M., Dahle, C., et al. (2017). A new high-resolution model of non-tidal atmosphere and ocean mass variability for de-aliasing of satellite gravity observations: AOD1B RL06. *Geophysical Journal International*, 211(1), 263–269. https://doi.org/10.1093/gij/ggx302
- Dobslaw, H., Dill, R., Bagge, M., Klemann, V., Boergens, E., Thomas, M., et al. (2020). Gravitationally consistent mean barystatic Sea Level rise from leakage-corrected monthly GRACE data. *Journal of Geophysical Research: Solid Earth*, 125(11), e2020JB020923. https://doi.org/10. 1029/2020jb020923
- Farrell, W. E., & Clark, J. A. (1976). On postglacial sea level. Geophysical Journal International, 46(3), 647–667. https://doi.org/10.1111/j.1365-246X.1976.tb01252.x
- Gałdyn, F., Sośnica, K., Zajdel, R., Meyer, U., & Jäggi, A. (2024). Long-term ice mass changes in Greenland and Antarctica derived from satellite laser ranging. *Remote Sensing of Environment*, 302, 113994. https://doi.org/10.1016/j.rse.2024.113994
- GRACE. (2018). GRACE_GSM_L2_GRAV_CSR_RL06. Ver. 6.0 [Dataset]. *PO.DAAC, CA, USA*. https://doi.org/10.5067/GRGSM-20C06 Gregory, J. M., Griffies, S. M., Hughes, C. W., Lowe, J. A., Church, J. A., Fukimori, I., et al. (2019). Concepts and terminology for Sea Level:

Mean, variability and change, both local and global. Surveys in Geophysics, 40(6), 1251–1289. https://doi.org/10.1007/s10712-019-09525-z Horwath, M., Gutknecht, B. D., Cazenave, A., Palanisamy, H. K., Marti, F., Marzeion, B., et al. (2022). Global sea-level budget and ocean-mass budget, with a focus on advanced data products and uncertainty characterisation. Earth System Science Data, 14(2), 411–447. https://doi.org/

- budget, with a focus on advanced data products and uncertainty characterisation. Earth System Science Data, 14(2), 411–44/. https://doi.org/ 10.5194/essd-14-411-2022
 Hsu, C. W., & Velicogna, I. (2017). Detection of sea level fingerprints derived from GRACE gravity data. Geophysical Research Letters, 44(17),
- Hsu, C. W., & Velicogna, I. (2017). Detection of sea level fingerprints derived from GRACE gravity data. *Geophysical Research Letters*, 44(17), 8953–8961. https://doi.org/10.1002/2017gl074070
- Jeon, T., Seo, K.-W., Kim, B.-H., Kim, J.-S., Chen, J., & Wilson, C. R. (2021). Sea level fingerprints and regional sea level change. *Earth and Planetary Science Letters*, 567, 116985. https://doi.org/10.1016/j.epsl.2021.116985
- Jiao, J., Pan, Y., Bilker-Koivula, M., Poutanen, M., & Ding, H. (2022). Basin mass changes in Finland from GRACE: Validation and explanation. Journal of Geophysical Research: Solid Earth, 127(6), e2021JB023489. https://doi.org/10.1029/2021jb023489
- Johnson, G. C., Hosoda, S., Jayne, S. R., Oke, P. R., Riser, S. C., Roemmich, D., et al. (2022). Argo-two decades: Global oceanography, revolutionized. *Annual Review of Marine Science*, 14(1), 379–403. https://doi.org/10.1146/annurev-marine-022521-102008
- Johnson, G. C., Lumpkin, R., Atkinson, C., Biló, T., Boyer, T., Bringas, F., et al. (2023). Global Oceans. Bulletin of the American Meteorological Society, 104(9), S146–S206. https://doi.org/10.1175/bams-d-23-0076.2

- Kim, J. S., Seo, K. W., Chen, J., & Wilson, C. (2022). Uncertainty in GRACE/GRACE-follow on global ocean mass change estimates due to mismodeled glacial isostatic adjustment and geocenter motion. *Scientific Reports*, 12(1), 6617. https://doi.org/10.1038/s41598-022-10628-8
- Kim, J. S., Seo, K. W., Jeon, T., Chen, J. L., & Wilson, C. R. (2019). Missing hydrological contribution to sea level rise. *Geophysical Research Letters*, 46(21), 12049–12055. https://doi.org/10.1029/2019g1085470
- Kvas, A., Brockmann, J. M., Krauss, S., Schubert, T., Gruber, T., Meyer, U., et al. (2021). GOCO06s—A satellite-only global gravity field model. *Earth System Science Data*, 13(1), 99–118. https://doi.org/10.5194/essd-13-99-2021
- Llovel, W., Balem, K., Tajouri, S., & Hochet, A. (2023). Cause of substantial global mean sea level rise over 2014–2016. Geophysical Research Letters, 50(19), e2023GL104709. https://doi.org/10.1029/2023gl104709
- Löcher, A., & Kusche, J. (2020). A hybrid approach for recovering high-resolution temporal gravity fields from satellite laser ranging. *Journal of Geodesy*, 95(1), 6. https://doi.org/10.1007/s00190-020-01460-x
- Long, D., Longuevergne, L., & Scanlon, B. R. (2015). Global analysis of approaches for deriving total water storage changes from GRACE satellites. Water Resources Research, 51(4), 2574–2594. https://doi.org/10.1002/2014wr016853
- Loomis, B. D., Rachlin, K. E., & Luthcke, S. B. (2019). Improved Earth oblateness rate reveals increased ice sheet losses and mass-driven sea level rise. *Geophysical Research Letters*, 46(12), 6910–6917. https://doi.org/10.1029/2019g1082929
- Loomis, B. D., Rachlin, K. E., Wiese, D. N., Landerer, F. W., & Luthcke, S. B. (2020). Replacing GRACE/GRACE-FO with satellite laser ranging: Impacts on Antarctic ice sheet mass change. *Geophysical Research Letters*, 47(3), e2019GL085488. https://doi.org/10.1029/ 2019g1085488
- Matsuo, K., Chao, B. F., Otsubo, T., & Heki, K. (2013). Accelerated ice mass depletion revealed by low-degree gravity field from satellite laser ranging: Greenland, 1991–2011. Geophysical Research Letters, 40(17), 4662–4667. https://doi.org/10.1002/grl.50900
- Meyer, U., Sosnica, K., Arnold, D., Dahle, C., Thaller, D., Dach, R., & Jäggi, A. (2019). SLR, GRACE and Swarm gravity field determination and combination. *Remote Sensing*, 11(8), 956. https://doi.org/10.3390/rs11080956
- Moreira, L., Cazenave, A., & Palanisamy, H. (2021). Influence of interannual variability in estimating the rate and acceleration of present-day global mean sea level. *Global and Planetary Change*, 199, 103450. https://doi.org/10.1016/j.gloplacha.2021.103450
- Nerem, R. S., Eanes, R. J., Thompson, P. F., & Chen, J. L. (2000). Observations of annual variations of the Earth's gravitational field using satellite laser ranging and geophysical models. *Geophysical Research Letters*, 27(12), 1783–1786. https://doi.org/10.1029/1999gl008440
- Nerem, R. S., & Wahr, J. (2011). Recent changes in the Earth's oblateness driven by Greenland and Antarctic ice mass loss. *Geophysical Research Letters*, 38(13), L13501. https://doi.org/10.1029/2011gl047879
- Pearlman, M., Arnold, D., Davis, M., Barlier, F., Biancale, R., Vasiliev, V., et al. (2019). Laser geodetic satellites: A high-accuracy scientific tool. Journal of Geodesy, 93(11), 2181–2194. https://doi.org/10.1007/s00190-019-01228-y
- Peltier, W. R., Argus, D. F., & Drummond, R. (2018). Comment on "An Assessment of the ICE-6G_C (VM5a) Glacial Isostatic Adjustment Model" by Purcell et al. Journal of Geophysical Research: Solid Earth, 123(2), 2019–2028. https://doi.org/10.1002/2016jb013844
- Rietbroek, R., Brunnabend, S. E., Kusche, J., Schroter, J., & Dahle, C. (2016). Revisiting the contemporary sea-level budget on global and regional scales. Proceedings of the National Academy of Sciences USA, 113(6), 1504–1509. https://doi.org/10.1073/pnas.1519132113
- Riva, R. E. M., Bamber, J. L., Lavallée, D. A., & Wouters, B. (2010). Sea-level fingerprint of continental water and ice mass change from GRACE. Geophysical Research Letters, 37(19), L19605. https://doi.org/10.1029/2010gl044770
- Scanlon, B. R., Zhang, Z., Rateb, A., Sun, A., Wiese, D., Save, H., et al. (2019). Tracking seasonal fluctuations in land water storage using global models and GRACE satellites. *Geophysical Research Letters*, 46(10), 5254–5264. https://doi.org/10.1029/2018gl081836
- Sośnica, K., Jäggi, A., Meyer, U., Thaller, D., Beutler, G., Arnold, D., & Dach, R. (2015). Time variable Earth's gravity field from SLR satellites. *Journal of Geodesy*, 89(10), 945–960. https://doi.org/10.1007/s00190-015-0825-1
- Storto, A., Masina, S., Balmaseda, M., Guinehut, S., Xue, Y., Szekely, T., et al. (2015). Steric sea level variability (1993–2010) in an ensemble of ocean reanalyses and objective analyses. *Climate Dynamics*, 49(3), 709–729. https://doi.org/10.1007/s00382-015-2554-9
- Storto, A., & Yang, C. (2024). Acceleration of the ocean warming from 1961 to 2022 unveiled by large-ensemble reanalyses. *Nature Communications*, 15(1), 545. https://doi.org/10.1038/s41467-024-44749-7
- Sun, Y., Riva, R., & Ditmar, P. (2016). Optimizing estimates of annual variations and trends in geocenter motion and J2 from a combination of GRACE data and geophysical models. *Journal of Geophysical Research: Solid Earth*, 121(11), 8352–8370. https://doi.org/10.1002/ 2016jb013073
- Swenson, S., Chambers, D., & Wahr, J. (2008). Estimating geocenter variations from a combination of GRACE and ocean model output. Journal of Geophysical Research, 113(B8), B08410. https://doi.org/10.1029/2007jb005338
- Swenson, S., & Wahr, J. (2002). Methods for inferring regional surface-mass anomalies from Gravity Recovery and Climate Experiment (GRACE) measurements of time-variable gravity. Journal of Geophysical Research, 107(B9), ETG-3. https://doi.org/10.1029/2001jb000576
- Talpe, M. J., Nerem, R. S., Forootan, E., Schmidt, M., Lemoine, F. G., Enderlin, E. M., & Landerer, F. W. (2017). Ice mass change in Greenland and Antarctica between 1993 and 2013 from satellite gravity measurements. *Journal of Geodesy*, 91(11), 1283–1298. https://doi.org/10.1007/ s00190-017-1025-y
- Tapley, B. D., Bettadpur, S., Ries, J. C., Thompson, P. F., & Watkins, M. M. (2004). GRACE measurements of mass variability in the Earth system. Science, 305(5683), 503–505. https://doi.org/10.1126/science.1099192
- Uebbing, B., Kusche, J., Rietbroek, R., & Landerer, F. W. (2019). Processing choices affect Ocean Mass estimates from GRACE. Journal of Geophysical Research: Oceans, 124(2), 1029–1044. https://doi.org/10.1029/2018jc014341
- Wahr, J., Molenaar, M., & Bryan, F. (1998). Time variability of the Earth's gravity field: Hydrological and oceanic effects and their possible detection using GRACE. Journal of Geophysical Research, 103(B12), 30205–30229. https://doi.org/10.1029/98jb02844

WCRP Global Sea Level Budget Group. (2018). Global sea-level budget 1993-present. Earth System Science Data, 10(3), 1551–1590. https://doi.org/10.5194/essd-10-1551-2018

- Willis, J. K., Chambers, D. P., & Nerem, R. S. (2008). Assessing the globally averaged sea level budget on seasonal to interannual timescales. Journal of Geophysical Research, 113(C6), C06015. https://doi.org/10.1029/2007jc004517
- Yi, S., Heki, K., & Qian, A. (2017). Acceleration in the global mean sea level rise: 2005–2015. *Geophysical Research Letters*, 44(23), 11905. https://doi.org/10.1002/2017g1076129
- Yi, S., Sun, W., Heki, K., & Qian, A. (2015). An increase in the rate of global mean sea level rise since 2010. *Geophysical Research Letters*, 42(10), 3998–4006. https://doi.org/10.1002/2015gl063902
- Yoder, C. F., Williams, J. G., Dickey, J. O., Schutz, B. E., Eanes, R. J., & Tapley, B. D. (1983). Secular variation of Earth's gravitational harmonic J2 coefficient from Lageos and nontidal acceleration of Earth rotation. *Nature*, 303(5920), 757–762. https://doi.org/10.1038/303757a0